

## **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1029/2018GL079605

#### **Key Points:**

- Large (5 microns) ash particles detected in the tropical lowermost stratosphere
- Ash particles were from the Mount Kelud eruption and persisted in the stratosphere for 4 weeks
- The low ash particle concentration indicates a limited impact on ice clouds or radiation

#### **Correspondence to:**

E. J. Jensen, eric.j.jensen@nasa.gov

#### Citation:

Jensen, E. J., Woods, S., Lawson, R. P., Bui, T. P., Pfister, L., Thornberry, T. D. et al. (2018). Ash particles detected in the tropical lower stratosphere. *Geophysical Research Letters*, *45*, 11,483–11,489. https://doi.org/10.1029/2018GL079605

Received 12 JUL 2018 Accepted 3 OCT 2018 Accepted article online 9 OCT 2018 Published online 29 OCT 2018

©2018. American Geophysical Union. All Rights Reserved. This article has been contributed to by US Government employees and their work is in the public domain in the USA.

# Ash Particles Detected in the Tropical Lower Stratosphere

Eric J. Jensen<sup>1</sup>, Sarah Woods<sup>2</sup>, R. Paul Lawson<sup>2</sup>, T. Paul Bui<sup>1</sup>, Leonhard Pfister<sup>1</sup>, Troy Dean Thornberry<sup>3,4</sup>, Andrew W. Rollins<sup>3</sup>, Jean-Paul Vernier<sup>5</sup>, Laura L. Pan<sup>6</sup>, Shawn Honomichl<sup>6</sup>, and Owen Brian Toon<sup>7</sup>

<sup>1</sup>NASA Ames Research Center, Moffett Field, CA, USA, <sup>2</sup>Spec, Inc., Boulder, CO, USA, <sup>3</sup>Chemical Sciences Division, NOAA Earth System Research Laboratory (ESRL), Boulder, CO, USA, <sup>4</sup>Cooperative Institute for Research in Environmental Science, Boulder, CO, USA, <sup>5</sup>NASA Langley Research Center, Hampton, VA, USA, <sup>6</sup>National Center for Atmospheric Research, Boulder, CO, USA, <sup>7</sup>Department of Atmospheric and Oceanic Sciences and Laboratory for Atmospheric and Space Physics, University of Colorado Boulder, Boulder, CO, USA

**Abstract** We report here on measurements of relatively large (up to  $\simeq$ 5-µm maximum dimension) particles detected in the tropical western Pacific lower stratosphere. The particles were observed at low relative humidities and contain no detectable ice mass. We conclude that these particles are silicate ash injected by the Mount Kelud eruption in Indonesia a few weeks prior to the aircraft observations. The Kelud ash particles were detected by cloud probes on the National Aeronautics and Space Administration Global Hawk between the lapse rate tropopause ( $\simeq$ 16.5 km) and the aircraft ceiling (18.5 km). The ash particles detected in the lowermost stratosphere by the aircraft extended well into the northern subtropics. The concentration of ash particles dropped off rapidly below the lapse rate tropopause, presumably because of rapid removal by ice nucleation scavenging. Although ash particles are very effective ice nuclei, the low particle concentrations detected in the upper troposphere precludes a significant impact on cirrus clouds.

**Plain Language Summary** We report here on measurements of relatively large (up to 5-µm maximum dimension) particles detected in the tropical western Pacific lower stratosphere during the National Aeronautics and Space Administration Airborne Tropical TRopopause EXperiment Global Hawk flights in early March 2014. The particles were observed at low relative humidities and contain no detectable ice mass. We conclude that these particles are silicate ash injected by the Mount Kelud volcanic eruption in Indonesia a few weeks prior to the Airborne Tropical TRopopause EXperiment flights. The Kelud ash particles were detected by cloud probes on the Global Hawk between the lapse rate tropopause (16.5 km) and the aircraft ceiling (18.5 km). The ash particles sampled in the lowermost stratosphere by the aircraft extended further north than the Kelud plume measured by spaceborne lidar at higher altitudes. The concentration of ash particles dropped off rapidly below the lapse rate tropopause, presumably because of rapid removal by ice nucleation on the ash particles followed by ice crystal sedimentation. Although ash particles are very effective ice nuclei, the low particle concentrations detected in the upper troposphere preclude a significant impact on cirrus microphysical properties.

## 1. Introduction

Explosive volcanoes that inject gas phase and particulate matter into the stratosphere have large impacts on interannual and decadal climate variability (Fyfe et al., 2013; Solomon et al., 2011). Most modeling studies focus on the volatile sulfate aerosols resulting from the volcanic sulfur injections, but silicate ash particles can also be injected into the stratosphere, and they can be an important component of the volcanic plume optical properties, at least during the first few weeks after the eruption (Oberbeck et al., 1983; Turco et al., 1983; Vernier et al., 2016). High-altitude measurements after the eruptions of Mount St. Helens in 1980 and El Chichon in 1982 showed that a broad size distribution of ash particles can be injected into the stratosphere, with particles as large as  $30-40 \mu m$  (Farlow et al., 1981; Gooding et al., 1983). The largest particles fall out of the stratosphere rapidly, but ash particles with maximum dimensions of a few microns persisted up to 6 months after the El Chichon eruption, possibly suggesting particle sedimentation speeds were slowed by irregular shapes and low-density clusters of particles (Gooding et al., 1983). Numerical simulations indicate





**Figure 1.** Cloud-Aerosol Lidar with Orthogonal Polarization (CALIOP) scattering ratio averaged over 3–10 March and 18–21 km. The location of the Mount Kelud eruption in Indonesia is indicated by the black triangle. The Airborne Tropical TRopopause EXperiment flight paths from Guam are shown in black for reference. In this altitude range, the Kelud stratospheric plume has spread throughout the tropics, but is mostly south of 5°N.

that particles larger than about  $5-\mu m$  diameter will be removed from the stratosphere by sedimentation within a few weeks (Turco et al., 1983).

As documented in detail by Vernier et al. (2016), the eruption of Mount Kelud (Indonesia) on 13 February 2014 resulted in a persistent aerosol plume in the tropical lower stratosphere. Backscatter and depolarization measurements from the Cloud-Aerosol Lidar with Orthogonal Polarization (CALIOP) on board Cloud-Aerosol Lidar and Infrared Pathfinder Satellite Observations satellite were used to investigate the evolution of the Kelud ash and sulfate particles in the lower stratosphere. The stratospheric aerosol plume spread throughout the global tropics within a few weeks (Figure 1), and lower stratosphere aerosol extinction remained well above background levels for at least a few months. Balloon-borne backscatter sondes launched from northern Australia in May 2014 indicated that fine ash (radii less than 0.3  $\mu$ m) persisted in the lower stratosphere 3 months after the eruption (Vernier et al., 2016). The sulfate aerosols and ash particles became vertically separated with time owing to faster sedimentation speeds of the larger ash particles. The sulfate aerosol layer remained between about 18 and 22 km, whereas the ash particles were mostly below 20 km. Note that the CALIOP analysis presented by Vernier et al. (2016) was restricted to altitudes above 18 km because cirrus occurring at lower altitudes can dominate the backscatter measurements.

During mid-February through early March 2014, the National Aeronautics and Space Administration Airborne Tropical TRopopause EXperiment (ATTREX) conducted flights from Guam with the long-range Global Hawk aircraft. The science foci of ATTREX were tropical tropopause layer (TTL;  $\simeq$ 14–18 km) transport, chemistry, cloud formation, and dehydration. The ATTREX Global Hawk payload provided measurements of water vapor, meteorological conditions, cloud properties, tracer and chemical radical concentrations, and radiative fluxes (Jensen et al., 2017). Although the ATTREX science team was aware of the Kelud eruption, flights were not specifically designed to target the aerosol plume because of a lack of aerosol instrumentation on the aircraft. However, as we will show in this paper, the Kelud ash particles in the lowermost stratosphere were detected by the cloud probes on the Global Hawk during the early March flights in the western Pacific Northern Hemisphere tropics. The in situ aircraft measurements provide information about particle size and concentration that is complementary to the CALIOP measurements.

#### 2. ATTREX Measurements

The key instrument used to detect the Kelud ash particles was the Fast Cloud Droplet Probe (FCDP; Lawson et al., 2017; McFarquhar et al., 2007; Woods et al., 2018), which measures the forward scattered light from individual particles. Mie scattering calculations are used to retrieve particle sizes from 1- to 50- $\mu$ m diameter. Without sufficient sulfate uptake, ash particles would be nonspherical, resulting in an FCDP sizing uncertainty of up to about a factor of 2 (Borrmann et al., 2000). The FCDP is an open-path instrument designed for sampling cloud droplets and ice crystals. The relatively low sample volume of FCDP (0.08 L/s; Thornberry et al., 2017) results in a minimum detectable particle concentration in 1-s samples of about 15 L<sup>-1</sup>. Since the lower stratospheric ash particle concentration was relatively low, we use 15- and 60-s averages of FCDP data in our analysis to obtain sufficient statistics for determination of particle size distributions and concentrations.





2014 ATTREX science flights

**Figure 2.** Airborne Tropical TRopopause EXperiment (ATTREX) February–March 2014 flight tracks from Guam. The thick sections of the flight tracks indicate where the aircraft was above the local lapse rate tropopause height (16.5 km on average; see text for discussion of tropopause height determination). Yellow dots along the flight paths indicate locations where 15-s samples indicated particles present (detected by Fast Cloud Droplet Probe), relative humidity with respect to ice < 50%, and no ice detected by NOAA-H<sub>2</sub>O. These particles were primarily detected when the aircraft was in the lowermost stratosphere during the last three flights (6–12 March).

The Global Hawk payload included two FCDP probes, one a stand-alone FCDP, and the other as part of the Hawkeye probe. The agreement between the ash particle properties indicated by the two FCDP instruments provides additional confidence that the particle detections are not instrument artifacts.

Important complementary measurements used to distinguish ash particles from ice crystals are temperature, pressure, water vapor concentration, and total water concentration. Temperature and pressure were measured by the Meteorological Measurement System, with a temperature uncertainty of 0.3 K. Water vapor was measured with both external-path and internal-path diode laser hygrometers. The two measurements showed excellent agreement even under the very dry conditions in the tropical lower stratosphere (Jensen et al., 2017). Here we use the internal-path NOAA-H<sub>2</sub>O water vapor measurement, which has an uncertainty of 5%  $\pm$ 0.23 ppmv (Thornberry et al., 2015). The resulting uncertainty in calculated relative humidity with respect to ice (RHI) is no more than 15%.

The NOAA-H<sub>2</sub>O instrument also includes a total water measurement using a forward-facing inlet (Thornberry et al., 2015). Ice crystals are inertially enhanced by a factor ranging from 33 to 48, depending on the particle size. The cloud ice water content (IWC) is obtained by differencing the total water and vapor channels, and the inertial enhancement provides a high sensitivity of about 2–3  $\mu$ g/m<sup>3</sup> (corresponding to about 0.02–0.03 ppmv at 100 hPa and 200 K; Thornberry et al., 2017). The ability to detect small amounts of ice present provides a sensitive discrimination between ash particle layers and ice clouds.

The six ATTREX Global Hawk flights from Guam covered the western Pacific region from about 10°S to about 30°N (Figure 2). The sampling strategy used most of the time was to profile between about 14 km and the Global Hawk ceiling, which varied between about 16 and 18.5 km, depending on the fuel load. This approach provided numerous vertical profiles through the TTL.

## 3. Observations of Ash Particles From Kelud Eruption

The central science focus of ATTREX was ice cloud processes controlling water vapor concentration in the TTL and in air entering the stratosphere. Examination of the RHI frequency distribution inside clouds is a useful method for evaluating numerical models of cirrus formation and interactions with water vapor. The RHI





**Figure 3.** Relative humidity frequency distributions during 15-s samples when Fast Cloud Droplet Probe (FCDP) detected particles. Black curve: All samples with FCDP particles; green curve: only including samples with condensed ice detected by the NOAA-H<sub>2</sub>O instrument. The mode at relative humidity with respect to ice (RHI) < 50% corresponds to the stratospheric ash particles.

frequency distribution including only 15-s samples where the FCDP probe detected particles (Figure 3) has the expected peak near 100% and tail extending to large supersaturations. However, a second peak is apparent at RHI less than 50%. This peak is almost exclusively produced by measurements made in the lower stratosphere. The possibility that these particles are sublimating ice crystals is implausible since small (diameters less than 10  $\mu$ m) ice crystals should sublimate within a few minutes at RHI < 50%. Further, if we include only samples with ice detected by the NOAA-H<sub>2</sub>O IWC measurement, then the peak at low RHI disappears (Figure 3, green curve), indicating that the particles detected at low relative humidity are not ice crystals. All of the observational data used in this study are publicly available on the Earth Science Project Office archive (espoarchive.nasa.gov).

Size distributions of the ash particles sampled in the lower stratosphere are shown in Figure 4. Size distributions from the two FCDP probes on the aircraft agree well. The particle maximum dimensions indicated by FCDP ranged from the lower detection limit (0.5  $\mu$ m) up to about 10  $\mu$ m; most of the particles had maximum dimensions less than about 5  $\mu$ m. These particles are much too large to be sulfate aerosols. The fall speed of a 3- $\mu$ m diameter spherical particle with a density of 2.5 g/cm<sup>3</sup> under lower strato-

spheric conditions is about 0.1 cm/s, which implies a sedimentation distance of about 3 km over the 3–4 weeks since the Kelud eruption. If the ash particles are nonspherical, they would fall at somewhat slower speeds. Likewise, if the ash particles are coated with sulfate, the density would be lower, resulting in slower sedimentation. During the first 10 days after the eruption, ash particles were detected by CALIOP in a primary layer between 18 and 20 km, with a secondary layer at about 22–23 km (Vernier et al., 2016). Therefore, it is plausible that the particles detected between  $\approx 16.5$  and 18.5 km during ATTREX were ash from the Kelud eruption.

The FCDP measurements also provide information about the lower stratospheric ash particle concentration. We use 60-s averages here to allow detection of low particle concentrations, but this involves averaging over about 10 km of horizontal distance, and we thereby lose information about variability in the particle concentrations at smaller scales. The FCDP measurements indicate lower stratospheric particle concentrations less than 10 L<sup>-1</sup>. Even with 10-km samples, the ash particle concentration is highly variable along the flight tracks (see Figure 2). The ash extinctions at visible wavelengths indicated by the FCDP measurements are



**Figure 4.** Stratospheric ash particle size distribution. Only samples above 16.5 km and with relative humidity with respect to ice (RHI) < 50% are included. The two curves show the size distributions indicated by the two Fast Cloud Droplet Probe (FCDP) probes on the Global Hawk flights from Guam.

about 0.0001–0.0003 km<sup>-1</sup>, which is comparable to the background aerosol extinction in the tropical lower stratosphere (Thomason & Vernier, 2013).

The relative humidities corresponding to samples with ash particles and ice crystals are plotted versus height relative to the lapse rate tropopause in Figure 5. For Global Hawk vertical profiles that extended into the lower stratosphere, the lapse rate tropopause is identified using a modified World Meteorological Organization definition (World Meteorological Organization, 1957), where we reduced the depth of the layer in which the average lapse rate must be 2 K/km from 2 to 0.5 km (Pan et al., 2018). For aircraft profiles that did not extend into the stratosphere, we use nearby (within 12 hr and 500 km) radiosonde or COSMIC profiles. The data from the few profiles with none of these options available are not included. Ash particles (red points) correspond to 15-s samples with FCDP detection of particles but no ice mass indicated by the NOAA-H<sub>2</sub>O IWC measurement. The green points show samples with ice detected (cirrus clouds). Occurrence frequency height distributions (solid curves in Figure 5) show that ice crystals were almost exclusively confined to heights below the lapse rate tropopause, whereas the ash was mostly detected above the tropopause. The concentrations of ash particles were correspondingly much lower in the upper troposphere than in the stratosphere. Also, Figure 5 shows





**Figure 5.** Height distributions of relative humidity with respect to ice relative to the lapse rate tropopause of 15-s samples with ice crystals (green circles, defined as particles detected by FCDP and IWC detected by NOAA-H<sub>2</sub>O) and ash particles (red triangles, defined as particles detected by FCDP and no IWC). The solid lines show the fractions of 15-s samples with ice or ash particles detected. See text for discussion of lapse rate tropopause height determination. FCDP = Fast Cloud Droplet Probe; IWC = ice water content.

expected clear distinction between low relative humidities where the ash particles were detected above the tropopause and high relative humidities in the ice clouds below the tropopause.

Although the ash particle concentration drops off sharply below the tropopause, some ash particles are detected in dry regions in the upper troposphere. Removal of ash particles in the lower stratosphere is controlled by horizontal dispersion and sedimentation, with corresponding long lifetimes. In the upper



**Figure 6.** Latitudes and heights of trajectories launched from the Kelud eruption location and time. Colors indicate the time since the eruption. White bars indicate the trajectory launch locations. The rapid poleward transport of parcels below about 18.5 km is evident. The Cloud-Aerosol Lidar with Orthogonal Polarization observations that were limited to heights above 18 km indicated ash particles only south of about 7°N (Figure 1), whereas ash particles were detected by the aircraft at altitudes below 18.5 km as far north as about 30°N (Figure 2).

troposphere, ash particles will be more rapidly removed by overturning circulations and ice nucleation scavenging (see discussion below). The difference in ash concentration between the lower stratosphere and upper troposphere may just be a result of the removal lifetime difference.

The CALIOP measurements at altitudes above 18 km indicated that the Kelud stratospheric aerosol enhancement was confined to latitudes south of 5°N in the ATTREX sampling time period and longitude range (Figure 1). In contrast, the ATTREX observations indicated the lower stratospheric ash layer extended well into the northern subtropics (Figure 2). However, the CALIOP analysis was confined to altitudes above 18 km. The more rapid poleward transport in the lowermost stratosphere is consistent with the known height variation in stratospheric latitudinal mixing and advection.

To understand the differences between the broad northward extent of the Kelud ash cloud at 16.5-18.5 km implied by Figure 2, and the latitudinally compact distribution above 18 km (Figure 1), we launched 30-day forward trajectories from a 3° by 3° region around Kelud at times bracketing the eruption at 16 Z, 13 February 2014. We used half-degree 6-hourly ERA-Interim analyses, carefully interpolated to very high vertical resolution in the TTL using the technique documented by Kim and Alexander (2013). Diabatic heating was radiative (including clouds), as determined by Yang et al. (2010). The results are in Figure 6, showing the positions of all parcels on 3-13 February inclusive, color coded by age. Above 18.5 km, the parcels are tightly confined, with some spread between 18 and 18.5 km.

The Northern Hemisphere TTL monsoon anticyclone near 17 km spreads some parcels northward as far as the subtropical jet. The Austral summer anticyclones spread parcels southward; the relatively weak summer sub-tropical jet is a less effective barrier to latitudinal transport than during wintertime. Note that there is some northward spread between 18 and 19 km. However, only about 3% of the parcel positions are north of 10°N in that altitude range (vs. 11% between 16.5 and 18 km).

We also examined data from specific CALIOP orbits in the ATTREX sampling region and time period. Aerosol enhancements in the lowermost stratosphere (between about 16.5 and 18 km) were not apparent. However, as noted above, the extinction associated with the ATTREX FCDP ash measurements is comparable to or lower than the background aerosol extinction; therefore, detection of these particles by CALIOP would not necessarily be expected, and the fact that the Kelud ash plume in the lowermost stratosphere is not detected by CALIOP does not indicate an inconsistency in the measurements. Essentially, the CALIOP and aircraft measurements provide complementary information about the Kelud plume: The aircraft measurements provide more sensitive detection of ash particles in the lowermost stratosphere than is possible with spaceborne lidar, whereas the CALIOP measurements provide global coverage over the entire evolution of the plume.

## 4. Summary and Discussion

We have presented measurements from the 2014 ATTREX Global Hawk flights in the tropical western Pacific that provided unexpected evidence of relatively large volcanic ash particles in the lowermost stratosphere. The key results are summarized as follows:

- 1. A widespread layer of relatively large (maximum dimensions up to about 5 μm) particles was detected in the tropical western Pacific lowermost stratosphere (between about 16.5 and 18.5 km) at relative humidities less than 50% on the ATTREX Global Hawk flights in early March 2014.
- 2. These particles contained no detectable ice mass and were presumably ash particles from the Mount Kelud eruption a few weeks earlier (13 February 2014). This conclusion is consistent with the CALIOP observations of the Kelud injection of ash well into the stratosphere as well as sedimentation speeds of 3-µm particles in the lower stratosphere. The aircraft measurements below 18.5 km indicate ash particles at higher northern latitudes than the CALIOP measurements above 18 km. This difference in latitudinal extent of the ash cloud at different altitudes is consistent with the height dependence in transport to the extratropics (Figure 6).
- 3. The measured concentrations of ash particles in the lower stratosphere are about  $1-10 L^{-1}$ . Extinctions calculated from the particle size and concentration measurements are comparable to the background aerosol extinction.
- 4. The concentration of particles that contain no detectable condensed water (i.e., not ice crystals) decreased sharply below the lapse rate tropopause (16.5 km on average). This change in concentration is presumably a result of rapid removal of the ash particles in the upper troposphere by nucleation scavenging and convective overturning.

Ash particles can be very effective ice nuclei (Schill et al., 2015). As a result, they can promote formation of cirrus clouds at lower relative humidities than is possible with only homogeneous nucleation or nucleation on less effective ice nuclei. However, the concentrations of ash particles in the lower stratosphere (no more than  $1-10 L^{-1}$ ) is relatively low, and the ash particle concentration drops rapidly below the tropopause where cirrus formation is possible. Cirrus with concentrations less than 10  $L^{-1}$  have very low optical depths, and growth of ice crystals in such low ice concentration cirrus will not appreciably deplete the vapor in excess of saturation (Jensen et al., 2013). In a steadily cooling air parcel, the relative humidity will continue to rise after nucleation on the few ash particles until additional ice crystals are generated by heterogeneous nucleation on more abundant, less effective ice nuclei (such as crystalline ammonium sulfate or glassy organic particles; Jensen et al., 2018) or by homogeneous freezing of aqueous aerosols. The resulting cloud microphysical and optical properties will be dominated by the less effective ice nuclei or homogeneous freezing. However, IN concentrations of 10 L<sup>-1</sup> have very little impact on the balance between heterogeneous and homogeneous ice nucleation. Significantly larger IN concentrations are required to alter cirrus microphysical properties. Overall, nucleation of ice crystals on the few Kelud ash particles present during the ATTREX flights probably did not significantly alter TTL cirrus microphysical properties or the water vapor budget. However, in the fresh ash plume just after the eruption, ash particle concentrations may have been high enough for significant impacts on cirrus.



Acknowledgments

This work was supported by NASA's Airborne Tropical Tropopause Experiment project. Data used in this study can be accessed at NASA Earth Sciences Project Office archive (espoarchive.nasa.gov).

#### References

- Borrmann, S., Luo, B., & Mischenko, M. (2000). Application of the T-matrix method to the measurement of aspherical (ellipsoidal) particles with forward scattering optical particle counters. *Journal of Aerosol Science*, *31*, 789–799.
- Farlow, N. H., Oberbeck, V. R., Stensinger, K. G., Ferry, G. V., Polkowski, G., & Hayes, D. M. (1981). Size distributions and mineralogy of ash particles in the stratosphere from eruptions of Mount St. Helens. *Science*, *211*, 832–834.
- Fyfe, J. C., von Salzen, K., Cole, J. N. S., Gillet, N. P., & Vernier, J.-P. (2013). Surface response to stratospheric aerosol changes in a coupled atmosphere-ocean model. *Geophysical Research Letters*, 40, 584–588. https://doi.org/10.1002/grl.50156
- Gooding, J. L., Clanton, U. S., Gabel, E. M., & Warren, J. L. (1983). El Chichon volcanic ash in the stratosphere: Particle abundances and size distributions after the 1982 eruption. Geophysical Research Letters, 10, 1033–1036.
- Jensen, E. J., Diskin, G., Lawson, R. P., Lance, S., Bui, T. P., Hlavka, D., et al. (2013). Ice nucleation and dehydration in the tropical tropopause layer. *Proceedings of the National Academy of Sciences of the United States of America*, *110*, 2041–2046. https://doi.org/10.1073/pnas.1217104110
- Jensen, E. J., Kärcher, B., Ueyama, R., Pfister, L., Bui, T. V., Diskin, G. S., et al. (2018). Heterogeneous ice nucleation in the tropical tropopause layer. *Journal of Geophysical Research: Atmospheres, 123*. https://doi.org/10.1029/2018JD028949
- Jensen, E. J., Pfister, L., Jordan, D. E., Bui, T. V., Ueyama, R., & Singh, H. B. (2017). The NASA Airborne Tropical TRopopause EXperiment (ATTREX): High-altitude aircraft measurements in the tropical western Pacific. *Bulletin of the American Meteorological Society*, *1*, 129–143. https://doi.org/10.1175/BAMS-D-14-00263.1
- Kim, J.-E., & Alexander, M. J. (2013). A new wave scheme for trajectory simulations of stratospheric water vapor. Geophysical Research Letters, 13, 5286–5290. https://doi.org/10.1002/grl.50963

Lawson, R. P., Gurganus, C., Woods, S., & Bruitjes, R. (2017). Aircraft observations of cumulus microphysics ranging from the tropics to midlatitudes: Implications for a "new" secondary ice process. *Journal of the Atmospheric Sciences*, 74, 2899–2920.

McFarquhar, G. M., Um, J., Freer, M., Baumgardner, D., Kok, G. L., & Mace, G. (2007). The importance of small ice crystals to cirrus properties: Observations from the Tropical Warm Pool International Cloud Experiment (TWP-ICE). *Geophysical Research Letters*, 57, L13803. https://doi.org/10.1029/2007GL029865

Oberbeck, V. R., Danielsen, E. F., Snetsinger, K. G., & Ferry, G. V. (1983). Effect of the eruption of El Chichon on stratospheric aerosol size and composition. *Geophysical Research Letters*, 10, 1021–1024.

Pan, L. L., Honomichl, S. B., Bui, T. V., Thornberry, T., Rollins, A., Hintsa, E., & Jensen, E. J. (2018). Lapse rate or cold point: The tropical tropopause identified by in situ trace gas measurements. *Geophysical Research Letters*, 45. https://doi.org/10.1029/2018GL079573

Schill, G. P., Genareau, K., & Tolbert, M. A. (2015). Deposition and immersion-mode nucleation of ice by three distinct samples of volcanic ash. Atmospheric Chemistry and Physics, 15, 7523–7536.

Solomon, S., Daniel, J., Neely, R. R. III, Vernier, J. P., Dutton, E. G., & Thomason, L. W. (2011). The persistently variable "background" stratospheric aerosol layer and global climate change. *Science*, 328, 866–870. https://doi.org/10.1126/science.1206027

- Thomason, L. W., & Vernier, J.-P. (2013). Improved SAGE II cloud/aerosol categorization and observations of the Asian tropopause aerosol layer: 1989–2005. Atmospheric Chemistry and Physics, 13, 4605–4616.
- Thornberry, T. D., Rollins, A. W., Avery, M. A., Woods, S., Lawson, R. P., Bui, T. V., & Gao, R. S. (2017). Ice water content-extinction relationships and effective diameter for TTL cirrus derived from in situ measurements during ATTREX 2014. *Journal of Geophysical Research: Atmospheres, 122,* 4494–4507. https://doi.org/10.1002/2016JD025948
- Thornberry, T. D., Rollins, A. W., Gao, R. S., Watts, L. A., Ciciora, S. J., McLaughlin, R. J., & Fahey, D. W. (2015). A two-channel, tunable diode laser-based hygrometer for measurement of water vapor and cirrus cloud ice water content in the upper troposphere and lower stratosphere. *Atmospheric Measurement Techniques*, *8*, 211–244.
- Turco, R. P., Toon, O. B., Whitten, R. C., Hamill, P., & Keesee, R. G. (1983). The 1980 eruptions of Mount St. Helens: Physical and chemical processes in the stratospheric clouds. *Journal of Geophysical Research*, 88, 5299–5319.
- Vernier, J.-P., Fairlie, T. D., Deshler, T., Natarajan, M., Knepp, T., Wienhold, F. G., et al. (2016). In situ and space-based observations of the Kelud volcanic plume: The persistence of ash in the lower stratosphere. *Journal of Geophysical Research: Atmospheres, 121*, 11,104–11,118. https://doi.org/10.1002/2016JD025344
- Woods, S., Lawson, R. P. R. P., Jensen, E. J., Bui, P., Thornberry, T., Rollins, A., et al. (2018). Microphysical properties of tropical tropopause layer (TTL) cirrus. Journal of Geophysical Research: Atmospheres, 123, 6053–6069. https://doi.org/10.1029/2017JD028068

World Meteorological Organization (1957). A three dimensional science: Second session of the commission for aerology. WMO Bulletin, IV(4), 134–138.

Yang, Q., Fu, Q., & Hu, Y. (2010). Radiative impacts of clouds in the tropical tropopause layer. *Journal of Geophysical Research*, *115*, D00H12. https://doi.org/10.1029/2009JD012393